Author’s response to reviews

Title: Association among house infestation index, dengue incidence, and sociodemographic indicators: Surveillance using geographic information system.

Authors:

Waldemir P Vargas (waldemir.vargas@ensp.fiocruz.br)
Hélia Kawa (heliakw@gmail.com)
Paulo C Sabroza (sabroza@ensp.fiocruz.br)
Valdenir B Soares (valdenir@ensp.fiocruz.br)
Nildimar A Honório (honorio@ioc.fiocruz.br)
Andréa S Almeida (andrea.almeida@ensp.fiocruz.br)

Version: 6 Date: 9 July 2015

Author’s response to reviews: see over
Dear Sr. Vargas Proel,

The following answer point-by-point the concerns of publishers.

1- They were added both requested items, "Competing Interests" and "Authors' Contributions" sections in the current version of the manuscript (07/09/2015), between “Conclusions and References”.

2- We revised manuscript conforms to the journal style.

With best wishes,

Andréa Sobral de Almeida